Author's response to reviews

Title: Association of CETP TaqI and APOE polymorphisms with Type II Diabetes mellitus in North Indians: a case control study

Authors:

Manjusha M Dixit (manjusha@sgpgi.ac.in)
Sandeep S Bhattacharya (physiology@sancharnet.in)
Balraj B Mittal (bml_pgi@yahoo.com)

Version: 3 Date: 27 June 2005

Author's response to reviews:

Editor, 
BMC Endocrine Disorders

Sub: Submission of a revised MS: 4511370715991503 entitled "Association of CETP TaqI and APOE polymorphisms with type II diabetes mellitus in North Indians: a case control study".

Dear Sir,

We are submitting revised MS: 4511370715991503 entitled "Association of CETP TaqI and APOE polymorphisms with type II diabetes mellitus in North Indians: a case control study".

Required corrections have been made and are as follows

1. Discretionary Revisions
   (i) Results, p8, para 1: p-value shown as (p=0.000) has been changed to (p=0.00008)
   (ii) Table 6: footnote has been removed
   (iii) Discussion para 5 (p12). "Studies in Chinese population shows enormous variation (0.2%-5%) [45, 46, 47, 48]. Which may be due to different ethnic groups studied; same is true for studies in other countries." has been changed to "Frequency of the D442G substitution in Chinese showed enormous variations (0.2%-5%) [45, 46, 47, 48] which may be due to ethnic differences in the studied populations."

2. Title has been changed using sentence case instead of capital letters, both in the manuscript file and the submission system.

3. Brackets round India in the affiliation details has been removed

4. Introduction has been renamed as 'Background' and Materials and methods as 'Methods'.

5. Competing interests and Authors' contributions sections has been inserted

6. Authors' qualifications have been deleted from the submission system.

7. After going through the manuscript spelling and grammatical mistakes and formatting have also been corrected.

Please acknowledge the receipt of the manuscript.

Yours Sincerely

Prof. Balraj Mittal